Author's response to reviews

Title: Support groups for dementia caregivers - Predictors for utilisation and expected quality from a family caregiver's point of view: A questionnaire survey

PART I*

Authors:

Elmar Graessel (elmar.graessel@uk-erlangen.de)
Angelika Trilling (angelika.trilling@stadt-kassel.de)
Carolin Donath (carolin.donath@uk-erlangen.de)
Katharina Luttenberger (katharina.luttenberger@uk-erlangen.de)

Version: 3 Date: 10 June 2010

Author's response to reviews:

We have realized all comments from the editor in the revised manuscript and highlighted the changes.

(Indication of Part I with referring to Part II, two typographical errors, some minor changes according to the formatting checklist (for example the position of the competing interests))